Author's response to reviews

Title: Between-airport heterogeneity in air toxics emissions associated with individual cancer risk thresholds and population risks

Authors:

Ying Zhou (ying.zhou@emory.edu)
Jonathan I Levy (jilevy@hsph.harvard.edu)

Version: 7 Date: 1 May 2009

Author's response to reviews:

Dear Environmental Health Editor:

A few minor formatting changes were made to Table 2 and 3 in the manuscript. Please let me know if you need additional information from me. Thank you very much for your help.

Best regards,
Ying Zhou